

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Han, Martijn A.

Working Paper Short-term managerial contracts and cartels

SFB 649 Discussion Paper, No. 2012-057

Provided in Cooperation with: Collaborative Research Center 649: Economic Risk, Humboldt University Berlin

Suggested Citation: Han, Martijn A. (2012) : Short-term managerial contracts and cartels, SFB 649 Discussion Paper, No. 2012-057, Humboldt University of Berlin, Collaborative Research Center 649 - Economic Risk, Berlin

This Version is available at: https://hdl.handle.net/10419/79633

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Short-Term Managerial Contracts and Cartels

Martijn A. Han *



* Humboldt-Universität zu Berlin, Germany

This research was supported by the Deutsche Forschungsgemeinschaft through the SFB 649 "Economic Risk".

http://sfb649.wiwi.hu-berlin.de ISSN 1860-5664

SFB 649, Humboldt-Universität zu Berlin Spandauer Straße 1, D-10178 Berlin



Short-Term Managerial Contracts and Cartels^{*}

Martijn A. Han[†]

October 2, 2012

Abstract

This paper shows how a series of commonly observed short-term CEO employment contracts can improve cartel stability compared to a long-term employment contract. When a manager's short-term appointment is renewed if and only if the firm hits a certain profit target, then (i) defection from collusion results in superior firm performance, thus reducing the chance of being fired, while (ii) future punishment results in inferior firm performance, thus increasing the chance of being fired in the future. The introduction of this re-employment tradeoff intertwines with the usual monetary tradeoff and can improve cartel stability. Studying the impact of fixed versus variable salary components, I find that fixed components can facilitate collusion with a short-term contract, while not affecting cartel stability with a long-term contract. Moreover, an extension of the model shows that short-term, renewable contracts can be a source of cyclical collusive pricing. Finally, interpreting the results in light of firm financing suggests that debt-financed firms can form more-stable cartels than equity-financed firms.

Keywords: cartels, collusion, managerial contracts, price wars JEL codes: L13, L22, L41

[†]Institute for Microeconomic Theory, Humboldt-Universität zu Berlin. Spandauer Strasse 1, D-10178 Berlin, Germany. Email: martijn.alexander.han@hu-berlin.de.

^{*}This paper is based on Chapter 3 of my PhD thesis Vertical Relations in Cartel Theory, written at the Amsterdam Center for Law & Economics (ACLE), University of Amsterdam; the research was initiated during a visit at GREMAQ, Toulouse School of Economics. I am grateful to both institutions. I thank Charles Angelucci, Arnoud Boot, Benno Bühler, Henrik Lando, Evgenia Motchenkova, Erik Pot, Patrick Rey, Julien Sauvagnat, Maarten Pieter Schinkel, Randolph Sloof, Lyn Tjon Soei Len, Jan Tuinstra, Jeroen van de Ven, Frank Verboven, Michael Yang, and especially Bruno Jullien, for constructive discussions and comments. I am also grateful to participants at CRESSE 2011 in Rhodos, AERIE 2010 in Istanbul, CLEEN 2010 in Amsterdam, NAKE 2010 in Utrecht, PREBEM 2010 at Nyenrode University in Breukelen, as well as to seminar participants at DIW in Berlin, DICE in Düsseldorf, and the Toulouse School of Economics. This research is supported by the Deutsche Forschungsgemeinschaft via the Collaborative Research Center 649 "Economic Risk".

1 Introduction

The majority of the literature on cartels and collusion treats the firm as a profit-maximizing decision maker, thereby deriving key insights on cartel stability and behavior from a profit-maximizing perspective. However, cartels often involve firms having separated ownership and control in such a way that the incentives of the key decision maker (CEO, manager) are not fully aligned with those of the profit-maximizing shareholders. The resulting corporate governance factors can have a key impact on the operation of cartels.¹

This paper studies how short-term, renewable employment contracts affect cartel stability and behavior. Although one may expect that short-sighted CEOs are incapable of running a stable cartel,² I show how *short-termism* generated through such contracts can in fact improve cartel stability. Moreover, contrary to conventional wisdom that profit-dependent remuneration can incentivize CEOs to misbehave, the model shows that short-term contracts with fixed (as opposed to: variable) salary components can improve cartel stability.

From a sample of 375 CEO employment contracts of large public corporations, Schwab and Thomas (2006) find that 81% specify a length of five years or less, with a mean of 3.6 years. Similarly, Gillian, Hartzell and Parrino (2009) find that the median and mean length in a sample of 184 S&P 500 CEO contracts is 3.4 and 3.0 years, respectively. Moreover, recent work by Jenter and Lewellen (2010) indicates that CEO turnover decisions are very sensitive to stock price performance: from a sample of 2,569 publicly traded U.S. firms from 1992 to 2005, they find that 44% of CEOs left the firm within 4 years as a result of bad firm performance.³ These empirical findings show that executives typically have a *shortterm, renewable employment contract*, that is, a contract for a finite period of time which is renewed if and only if the firm performs sufficiently well.

In this paper, I take such a typical CEO contract as exogenously given and aim to understand its impact on the operation of cartels.⁴ Assuming that managerial remuneration as well as the probability of contract renewal is positively related to firm profit, the manager can stochastically increase profit by forming a cartel with rival firms, while even further stochastically increasing profit by defecting from the collusive agreement. Two interrelated forces then dictate cartel stability. The manager compares the immediate expected salary gain resulting from defection, with the future expected salary losses resulting from punishment

¹See Buccirossi and Spagnolo (2007) for a research agenda on corporate governance factors and collusion. ²For example, Aubert, Rey and Kovacic (2006) argue that short-term contracts may incentivize managers to apply for leniency as they may leave the firm anyway; this effect tends to destabilize collusion.

³For additional evidence on the relationship of CEO turnover and firm performance, see Lausten (2002), Farrell and Whidbee (2003), Jenter and Kanaan (2010), and Kaplan and Minton (2010).

⁴The model abstracts away from the underlying motivation of adopting short-term, renewable contracts over long-term contracts or no contract at all. Such motivations include risk sharing or adverse selection—see the papers referenced above.

(monetary tradeoff). The manager also compares the immediate increase in the probability of contract renewal resulting from the higher expected profit during defection, with the future decrease in the probability of contract renewal resulting from the lower expected profits during future punishment (re-employment tradeoff). While the monetary tradeoff parallels the familiar problem faced by integrated, profit-maximizing firms engaging in cartelization, the re-employment tradeoff originates from the manager's fear of not being rehired.

Whether a series of short-term, renewable contracts strictly enhance cartel stability compared to a long-term contract depends on two effects. First, the manager is unsure whether he will be rehired even if he colludes and, hence, the game ends for him with some positive probability: this is a collusion-destabilizing effect. Second, noting that after defection the manager only cares about re-employment if the expected remuneration during punishment is not too small, the re-employment tradeoff can dominate the collusion-destabilizing effect if the probability of re-employment is relatively low during punishment, while the expected remuneration during punishment is indeed not too small. This ensures that the manager misses out (in expectation) on a substantial amount of future salary as he is fired in the future with a high probability.

The model also shows how a fixed salary component can stabilize collusion when a shortterm contract is in place, while not affecting cartel stability when a long-term contract is in place. The intuition is that with a long-term contract, the fixed salary component is paid out in every period independent on managerial behavior and, therefore, does not affect the manager's incentives to collude. However, with a short-term contract, the fixed component is only paid out in periods in which the manager is indeed working for the firm. The manager anticipates that defection leads to a higher probability of being fired in the future—that is, losing out on future fixed salary components—which may amount to a large expected loss when he is relatively patient. In that case, the board of shareholders (more generally: the principal) optimally makes the fixed component an important part of the manager's salary so as to stabilize collusion.

In addition, this paper studies the case in which managers may revert from punishment to collusion when one of the managers is fired and replaced by another. The rationale behind such "serial collusion" is that managerial replacement restores the trust needed for a cartel to work. Such restoration of trust destabilizes collusion. The reason is that serial collusion increases the continuation value of defection, because (i) defection instantaneously increases the probability that the rival manager is fired, in which case collusion is immediately restored, and (ii) punishment is less fierce as it ends when the rival manager is fired at some point.

Finally, I consider the case in which managers engage in multiple sequential interactions within the span of the employment contract, thereby showing that short-term employment contracts may be an explanation for cyclical collusive pricing. Suppose, for example, that the manager is halfway through his employment contract and has been colluding up until now, but that earlier firm profits have been extremely low due to random market conditions. In such a case, the profits during the second half of his employment contract should be extremely high for the manager's contract to be renewed. Since extremely high profits are very unlikely to occur even with collusion, the manager anticipates that his contract is unlikely to be renewed, resulting in defection from the collusive agreement. However, rational managers fix such destabilization by *ex ante* agreeing that the collusive strategy entails collusion if and only if earlier firm profits during the same employment contract have not been too low. The optimal collusive strategy then entails waves of competition, i.e., price wars, in equilibrium.

This paper is organized as follows. Section 2 discusses related literature. Section 3 presents the model, identifies the key tradeoff, and shows how short-term contracts can increase cartel stability. Section 4 investigates the impact of fixed salary components and serial collusion on cartel stability. Section 5 considers an employment contract covering multiple interactions on the product market over the contractual period, thereby allowing for cyclical collusive pricing. Section 6 reinterpret the results in light of firm financing, thereby arguing how debt-financed firms can form more-stable cartels than equity-financed firms. Section 7 concludes.

2 Related Literature

This paper relates to two strands of literature: (i) managerial incentives in cartels, and (ii) collusion and price wars. I discuss how my work relates to each in turn.

Managerial incentives in cartels. This paper departs from the traditional literature on cartels and collusion by separating the firm's ownership and control so as to focus on the impact of short-term contracts on cartel stability and behavior. The papers discussed hereafter deal with the impact of several other common characteristics of managerial compensation on cartels.⁵

Spagnolo (2000) shows that stock-related compensation improves the stability of collusion: if stock markets have perfect foresight and are informed about the collusive agreement, then future punishment following a defection is immediately discounted in current stock prices,

⁵The strategic delegation literature considers a related issue of how delegation affects the competitive outcome on the product market. The seminal papers are Fershtman and Judd (1987) and Sklivas (1987), which were preceded by Fershtman (1985) and Vickers (1985). Lambertini and Trombetta (2002), De Lamirande, Guigou and Lovat (2011), and Han (2011) extend the seminal strategic delegation models to a repeated setting so as to allow for collusion.

thereby instantaneously reducing the gain from defection.⁶ Moreover, Spagnolo (2005) argues that collusion is more stable when managers have a preference for income smoothing, that is, when managers are provided with low-powered incentives so that they prefer a smooth stream of profits over time. The manager is then averse to variance in profits, which disincentivizes him to defect as that leads to a high profit today, followed by low punishment profits in the future. Also, Spagnolo (2005) shows that well-chosen capped bonus plans allow collusion to be stable for any discount factor, because the gain from defection will be capped. In contrast to Spagnolo's work, this paper specifically allows contracts to be finite, thereby introducing a collusion-stabilizing tradeoff based on the manager's fear to be fired.

Since cartels are illegal, a written contract between the firm owner and the manager that explicitly induces the manager to form a cartel is not enforceable in court. Therefore, the firm owner needs to rely on a relational contract with the manager to induce cartelization. Chen (2008) shows how such a relational contract between the owner and the manager facilitates collusion as (i) it prevents the manager to cheat on the product market, because defection results in collusion breaking down in which case the owner will pay nothing to the manager and the relational contract breaks down, while (ii) the owner cannot cheat on the relational contract by inducing the manager to defect, because the owner cannot credibly commit to rewarding the manager for a defection. Thus, Chen (2008) concludes that there exists no selfenforcing relational contract that can induce the manager to cheat on the product market. This paper, however, circumvents the need for a relational contract as contracting on profits in a short-term contract introduces a new tradeoff that can be used to stabilize collusion.

The manager's decision to form a cartel may interfere with other managerial duties. Aubert (2009) builds an elegant model of double moral hazard in which the manager chooses whether to exert costly effort which increases profits, while either competing or colluding on the product market. In such a scenario, a high-powered incentive contract aimed at inducing the manager to exert effort has the perverse effect of also inducing the manager to form a cartel. This conflict between incentives results in equilibrium contracts inducing a suboptimal effort level, leading to welfare losses even if there is no collusion in equilibrium. In a similar model of (single) moral hazard, Angelucci and Han (2011) consider a three-tier hierarchy, *authority-shareholder-manager*, so as to study the effects of intra-firm monitoring on fighting corporate crime, such as cartels. While Aubert (2009) and Angelucci and Han (2011) take a general approach to open up the black box of the profit-maximizing firm so as to endogenously determine the optimal contract, this paper takes the short-termism and renewability of observed CEO contracts as exogenously given.

 $^{^{6}}$ In a market with stochastic, autocorrelated demand development, Neubecker (2005) confirms Spagnolo's (2000) finding that stock-based compensation facilitates collusion.

More generally, Bull (1987) and Baker, Gibbons and Murphy (2002), for instance, study relational contracts. They argue that such informal agreements and unwritten codes of conduct are widespread within firms and have an important impact on the incentives of individuals. Baker, Gibbons and Murphy (2002) show that the optimal allocation of decision rights minimizes the maximum temptation to renege on relational contracts. This paper takes the perspective of common formal, short-term, renewable CEO contracts.

Cartels and price wars. Empirical work shows that episodes of collusion often involve price wars—see, for example, Slade (1990), Harrington (2006), and Levenstein and Suslow (2006).⁷ As a founder of cartel theory, Stigler (1964) pointed out that prices are high during cartelization, while *out-of-equilibrium* price wars occur after cartel breakdown. Since then, many authors have contributed to explaining patterns of elevated prices and price wars *in equilibrium*.

Such explanations include imperfect monitoring with demand uncertainty (Green and Porter, 1984; Abreu, Pearce and Stacchetti, 1986), demand cyclicity (Rotemberg and Saloner, 1986; Haltiwanger and Harrington, 1991), learning about unknown demand parameters with exogenous demand shocks (Slade, 1989), renegotiation of the collusive pie (Levenstein, 1997), and inducing exit of "dying" cartel members (Fershtman and Pakes, 2000).⁸ These contributions treat the firm as profit-maximizing integrated entities.

In contrast, this paper presents an explanation for equilibrium price wars based on managerial issues. Since executives' compensation depends on their performance during the *full* contractual employment period, they optimally condition their pricing decision on profits realized earlier during their contractual employment period.

3 A Model of Cartel Stability with Short-Term Contracts

This section introduces the model (3.1); identifies the tradeoff associated with short-term contracts and cartel stability (3.2); shows that short-term contracts can improve cartel stability (3.3); and performs comparative statics of the key proposition (3.4).

⁷I follow the literature by using the term "price wars" for episodes of lower prices than the collusive price, which can be caused by cartel break-down or endogenously arising as part of the collusive equilibrium.

⁸More recent contributions include the cartel members' concerns about detection by the antitrust authority (Harrington, 2004a), lifting prices after detection to push for lower damages in court (Harrington, 2004b), and cost variability in the presence of buyer detection (Harrington and Chen, 2006).

3.1 Set-up of the Model

Outline & players. Consider two firms $i \in \{1, 2\}$, where each firm i is owned by shareholder i who employs a manager i to run the firm. In an infinitely repeated game, the managers interact with each other on the product market in each period $t \in \{1, ..., \infty\}$. I refer to shareholders in the female form (she/her), managers in the male form (he/his) and firms in the neutral form (it/its). Figure 1 graphically summarizes the players.



Figure 1. The players: shareholders and managers.

Managerial action. In each period $t \in \{1, ..., \infty\}$, both managers simultaneously decide whether to compete or collude on the product market. Managerial behavior stochastically affects the realization of the firms' profits.⁹ Let profit of firm *i* in period *t* be a random variable $\pi_t \ge 0$ with cumulative distribution function (CDF)

$$F_{a}(x) = \Pr\left[\pi_{t} \leq x | a\right], \forall x \geq 0,$$

where $a \in \{N, C, D, T\}$ represents the combination of behavior of manager *i* and *j* on the product market,

- 1. a = N if both managers compete (Nash competition);
- 2. a = C if both managers collude (Collusion);
- 3. a = D if manager *i* competes and manager *j* colludes (Defection by manager *i*);
- 4. a = T if manager *i* colludes and manager *j* competes (manager *i* has been <u>Tricked</u>, because manager *j* defected).

⁹Gillian, Hartzell and Parrino's (2009) empirical results show that an explicit short-term contract is more likely to be implemented when there is greater uncertainty about firm performance.

Assume that $F_a(x)$ is increasing over the entire domain $x \ge 0$, i.e., $\frac{\partial F_a(x)}{\partial x} > 0$ has full support. Defection is more profitable in expectation than collusion, which is, in turn, more profitable in expectation than Nash competition, in the sense of first order stochastic dominance, that is,

$$F_D(x) \le F_C(x) \le F_N(x),$$

for every $x \ge 0$, with a strict inequality for at least one value of x. By definition of stochastic dominance, expected profit in period t, $E_a(\pi_t)$, then satisfies $E_D(\pi_t) > E_C(\pi_t) > E_N(\pi_t)$. Moreover, collusion stochastically dominates being tricked, that is,

$$F_C(x) \le F_T(x), \tag{1}$$

for every $x \ge 0$, with a strict inequality for at least one value of x. Thus, expected profits are such that $E_C(\pi_t) > E_T(\pi_t)$.

To stay in line with the literature and to focus on the impact of short-term contracts on cartel stability, both managers adopt the grim-trigger collusive strategy: a manager colludes as long as his rival colluded in all previous periods, but reverts to Nash forever after a defection.

Shareholder's actions. The shareholder appoints her manager for one period by offering a contract that is renewed for another period if and only if the firm's profit π_t is above some threshold profit level $\underline{\pi} \in [0, \infty)$ chosen by the shareholder in period t = 0. Otherwise, the manager is fired and replaced by a new manager. This simple rule of *perform well or beat it* is motivated by (i) observations of CEOs and managers being fired after bad firm performance;¹⁰ (ii) the debate in academic journals as well as in the popular press that shareholders are focused on the short-term;¹¹ and (iii) the fact that the stock market allows investors (shareholders) to invest in a variety of financial products, thus rationalizing the shareholder to require a minimum threshold profit level that at least *beats the stock market*.¹²

In period t = 0, the shareholder also sets the managerial per-period salary scheme, consisting of a fraction $\beta \in [0, 1]$ of per-period profit π_t , plus a fixed component $\alpha \ge 0$ in each period.^{13,14}

¹⁰See, for example, Lausten (2002), Farrell and Whidbee (2003), and Jenter and Kanaan (2010).

¹¹See, for example, Bolton, Scheinkman and Xiong (2006).

¹²Kaplan and Minton (2010) make the specific point that CEO replacement is related to firm performance relative to the performance of the overall stock market.

¹³This paper does not consider shareholders colluding in setting managerial incentive schemes. See Lambertini and Trombetta (2002) and Han (2011) for models in which firm owners collude on this.

¹⁴To focus on the impact of short-term managerial contracts on cartel stability, the model abstract away from dynamic contracting and does not allow managerial compensation to be based on profit realizations in previous or future periods. Section 5 considers the contract to be contingent on *previous* profit realizations;

Information. Managerial behavior is observable to the rival manager, but unobservable to shareholders. The motivation is that close managerial interaction on the product market reveals their actions to each other. However, as cartels are illegal, it would be virtually impossible for managers to credibly communicate to shareholders that there is a cartel without creating suspicion from the law enforcers.

Payoffs and objectives. In each period, the manager receives his salary, while the shareholder receives realized firm profit minus managerial salary. In period t, the shareholder's net profit $\Pi_t(\pi_t)$ and the manager's salary $S_t(\pi_t)$ are, respectively,

$$\Pi_t (\pi_t) = (1 - \beta) \pi_t - \alpha_t$$
$$S_t (\pi_t) = \alpha + \beta \pi_t.$$

The manager has limited liability, which is taken care of by α and β being non-negative by assumption. The shareholder and the manager are risk neutral, have zero outside options, and discount payoffs with factor δ . The manager's objective is to maximize his discounted stream of expected payoffs. The owner's objective is to maximize cartel stability.¹⁵

Timing. In period t = 0, both shareholders independently set salary scheme (α, β) , as well as the threshold profit level $\underline{\pi}$ that a manager needs to realize to be reappointed. In all subsequent periods $t \in \{1, ..., \infty\}$, (i) the managers interact on the product market by taking action pair $a \in \{N, C, D, T\}$; (ii) firm profit π_t is realized; and (iii) the manager is either fired or reappointed, depending on the realization of profit and the profit threshold level $\underline{\pi}$ set in period t = 0. The timing of the game is graphically depicted in Figure 2.



Figure 2. Timing of the game.

for compensation schemes based on *future* profit realizations, see Spagnolo (2000), discussed in Section 2. ¹⁵It is left for future work to consider the owner to maximize her discounted stream of expected payoffs. The current model focuses on the impact of short-term contracts on cartel stability and behavior without fully endogenizing the employment contract.

3.2 Cartel Stability: Monetary and Re-employment Tradeoff

To focus on the main tradeoff, assume for now that managerial salary entails no fixed component, i.e., $\alpha = 0$. Subsection 4.1 relaxes this assumption.

As usual in the literature on collusion, the measure of cartel stability is the lowest discount factor δ such that the gain from defection does not exceed the expected discounted punishment: the lower is this discount factor, the more stable is the cartel.

Benchmark stability. Before solving the game outlined above, consider the benchmark model in which the manager has a long-term employment contract: he is hired forever without ever being fired.¹⁶ In that case, the necessary and sufficient condition for collusion to be stable is

$$\sum_{t=1}^{\infty} \delta^{t-1} \beta E_C(\pi_t) \ge \beta E_D(\pi_t) + \sum_{t=2}^{\infty} \delta^{t-1} \beta E_N(\pi_t),$$

iff.
$$\underbrace{\frac{\delta}{1-\delta}}_{A} \underbrace{\left(E_C(\pi_t) - E_N(\pi_t)\right)}_{B} \ge \underbrace{E_D(\pi_t) - E_C(\pi_t)}_{C},$$
(2)

which represents the familiar result that collusion is stable if and only if the immediate expected monetary gain from defection (C) is not larger than the expected monetary loss from punishment in each future period (B), discounted to the present (A). Fraction β does not turn up in the stability condition as it cancels out in each term.

Stability with short-term contracts. For notational convenience, denote by $G_a(x) = 1 - F_a(x) = \Pr[\pi_t \ge x | a]$ the probability of realizing profit $\pi_t \ge x$ when the managers take action combination $a \in \{N, C, D, T\}$. Thus, given that the shareholder requires a profit of at least $\underline{\pi}$, the probability of re-employment for another period is $G_a(\underline{\pi})$. The following lemma states the stability of collusion in the model with short-term contracts.

Lemma 1 With short-term contracts, collusion is stable if and only if

$$\delta\left(\frac{G_C\left(\underline{\pi}\right)}{1-\delta G_C\left(\underline{\pi}\right)}E_C\left(\pi_t\right) - \frac{G_D\left(\underline{\pi}\right)}{1-\delta G_N\left(\underline{\pi}\right)}E_N\left(\pi_t\right)\right) \ge E_D\left(\pi_t\right) - E_C\left(\pi_t\right). \tag{3}$$

¹⁶In reality, even if the manager has a long-term employment contract, the shareholders may find some way to fire the manager, for example, by creatively interpreting the *just cause* doctrine to terminate the contract or to settle on an appropriate "golden handshake" with the manager. To focus on the impact of contract duration on cartels, this paper abstracts away from such behavior and implicitly assume that the termination costs of a long-term contract are infinitely high.

Proof. Collusion is stable if and only if the discounted expected payoff from defection is not larger than the discounted expected payoff from collusion, i.e., if and only if

$$\sum_{t=1}^{\infty} \left(\delta G_C\left(\underline{\pi}\right)\right)^{t-1} \beta E_C\left(\pi_t\right) \ge \beta E_D\left(\pi_t\right) + \delta G_D\left(\underline{\pi}\right) \sum_{t=1}^{\infty} \left(\delta G_N\left(\underline{\pi}\right)\right)^{t-1} \beta E_N\left(\pi_t\right),$$

iff.
$$\frac{E_C\left(\pi_t\right)}{1 - \delta G_C\left(\underline{\pi}\right)} \ge E_D\left(\pi_t\right) + \delta G_D\left(\underline{\pi}\right) \frac{E_N\left(\pi_t\right)}{1 - \delta G_N\left(\underline{\pi}\right)},$$

which yields condition (3) when rearranging terms.

The stability condition in Lemma 1 compares the immediate expected gain from defection (RHS) with the discounted expected future losses from punishment (LHS). As is the case in the benchmark stability condition, fraction β cancels out in each term. Stability now depends on two interrelated tradeoffs: (i) the immediate expected monetary gain from defection, compared with the discounted expected monetary loss from future punishment (monetary tradeoff), and (ii) the immediate increase in re-employment probability from defection, compared to the future decrease in re-employment probability due to punishment (re-employment tradeoff).

The monetary tradeoff and the re-employment tradeoff are intertwined. To gain intuition, condition (3) can be rewritten as

$$\underbrace{\frac{G_{C}(\underline{\pi})\delta}{1-G_{C}(\underline{\pi})\delta}}_{C}\underbrace{(E_{C}(\pi_{t})-E_{N}(\pi_{t}))}_{B} + \underbrace{\left(\underbrace{\frac{G_{C}(\underline{\pi})\delta}{1-G_{C}(\underline{\pi})\delta} - \frac{G_{D}(\underline{\pi})\delta}{1-G_{N}(\underline{\pi})\delta}}_{E}\right)}_{E}\underbrace{E_{N}(\pi_{t})}_{D,E} \\
\geq \underbrace{E_{D}(\pi_{t}) - E_{C}(\pi_{t})}_{A},$$
(4)

which shows that collusion is stable if and only if

(A) the immediate expected monetary gain from defection is not larger than

(B) the future per-period expected monetary loss from punishment, taking into account

- (C) that the probability of re-employment is $G_C(\underline{\pi})$ in each equilibrium period, which
- (D) increases to $G_D(\underline{\pi})$ in the period of defection, while
- (E) decreasing to $G_N(\underline{\pi})$ in all following punishment periods.

Below, I describe the intuition behind these effects and how they compare to the effects in the benchmark model. (A) Monetary gain. The expected monetary gain from defection, $E_D(\pi_t) - E_C(\pi_t)$, is the same as in the benchmark model. The reason is that the immediate expected monetary gain is not contingent on whether the manager will be rehired in the future.

(B) Monetary loss. Provided that a punishment period is reached, the per-period expected monetary loss from punishment, $E_C(\pi_t) - E_N(\pi_t)$, is the same as in the benchmark model, because the expectations over profit realizations do not change by allowing managers to be fired.

(C) Continuation probability. In equilibrium, the probability that a manager is being reappointed is $G_C(\underline{\pi})$ in each period, whereas this probability is one in the benchmark model. When discounting the expected monetary losses from punishment (B) to the present, we should, therefore, premultiply the discount factor by $G_C(\underline{\pi})$. This decreases the manager's discounted expected loss of future punishment compared to the benchmark model and, thus, has a collusion-destabilizing effect.

(D) Re-employment gain. Defection results in a one-time *increase* in the probability of re-employment from $G_C(\underline{\pi})$ to $G_D(\underline{\pi})$. As a result, although the expected monetary payoff decreases to $E_N(\pi_t)$ after defection, the continuation probability of the game increases from $G_C(\underline{\pi})$ to $G_D(\underline{\pi})$. This effectively leads to a reduction in the expected loss from punishment, which is indicated by the numerators of the factors that premultiply $E_N(\pi_t)$ in condition (4). This reduction in the loss from punishment amounts to destabilizing collusion.

(E) Re-employment loss. Punishment leads to a *decrease* in the probability of reappointment from $G_C(\underline{\pi})$ to $G_N(\underline{\pi})$ in all future periods. Therefore, not only decreases the expected monetary payoff to $E_N(\pi)$, the continuation probability in all future periods of the game decreases from $G_C(\underline{\pi})$ to $G_N(\underline{\pi})$. This makes punishment fiercer through an increase in the expected loss from punishment, which is indicated by the denominators of the factors that premultiply $E_N(\pi_t)$ in condition (4). This increase in the loss from punishment amounts to stabilizing collusion.

3.3 Short-Term Contracts Can Increase Cartel Stability

Given profit threshold level $\underline{\pi}$, Proposition 1 states the characteristics of the probability distributions over profits $F_a(\cdot)$ such that collusion with short-term contracts is more stable than collusion with long-term contracts.

Proposition 1 Collusion with short-term contracts and threshold profit level $\underline{\pi}$ is more stable than collusion with long-term contracts if and only if

$$\frac{E_C(\pi_t)}{E_N(\pi_t)} < \frac{1 - \delta G_C(\underline{\pi})}{1 - \delta G_N(\underline{\pi})} \frac{1 - \delta G_N(\underline{\pi}) - (1 - \delta) G_D(\underline{\pi})}{1 - G_C(\underline{\pi})}.$$
(5)

Proof. Collusion is more stable compared to the benchmark model if and only if condition (3) is satisfied for more discount factors than condition (2), i.e., if and only if

$$\delta\left(\frac{G_{C}\left(\underline{\pi}\right)}{1-G_{C}\left(\underline{\pi}\right)\delta}E_{C}\left(\pi_{t}\right)-\frac{G_{D}\left(\underline{\pi}\right)}{1-G_{N}\left(\underline{\pi}\right)\delta}E_{N}\left(\pi_{t}\right)\right)>\frac{\delta\left(E_{C}\left(\pi_{t}\right)-E_{N}\left(\pi_{t}\right)\right)}{1-\delta},$$
(6)

which yields condition (5) when simplifying.

Short-term contracts stabilize collusion if and only if (i) the increase in reappointment probability during defection from $G_C(\underline{\pi})$ to $G_D(\underline{\pi})$ is relatively small compared to the reduction in reappointment probability during punishment $G_C(\underline{\pi})$ to $G_N(\underline{\pi})$ (RHS), while (ii) the monetary loss during punishment from $E_C(\pi_t)$ to $E_N(\pi_t)$ is not too large (LHS). The intuition is based on the following three forces.

First, with a short-term contract, the manager is *ex ante* unsure whether he will be rehired even if he colludes. This has a destabilizing impact on collusion as it incentivizes the manager to deviate so as to grab as much salary as possible in the current period, while at the same time increasing the immediate probability of re-employment.

Second, short-term contracts introduce the re-employment tradeoff, which has a stabilizing impact on collusion if and only if the immediate increase in re-employment probability resulting from defection is small compared to the future reduction in re-employment probability resulting from punishment, that is, if and only if $G_D(\underline{\pi}) - G_C(\underline{\pi})$ is small relative to $G_C(\underline{\pi}) - G_N(\underline{\pi})$. If the re-employment tradeoff is sufficiently positive for collusive stability, it can offset the first force.

Third, the higher is the reduction in expected salary resulting from punishment, the less does the manager care about being rehired in the future, that is, the lower is the impact of the re-employment tradeoff, i.e., the second force.

Altogether, the re-employment tradeoff (force 2) stabilizes collusion by more than force 1 destabilizes collusion if and only if (i) the increase in reappointment probability during defection is small compared to the decrease in reappointment probability during punishment, provided that (ii) the monetary loss from punishment is not too large. **Proposition 2** Short-term contracts can stabilize cartels compared to long-term contracts.

Proof. If condition (5) holds for some $\underline{\pi}$, then the shareholder chooses one of those $\underline{\pi}$ such that (3) is satisfied. If condition (5) does not hold for any $\underline{\pi}$, then the shareholder sets $\underline{\pi} = 0$, thereby effectively mimicking a long-term contract.

When there exists a profit level such that condition (6) is satisfied, then the optimal short-term contract stabilizes collusion. When no such profit level exists, the optimal short-term, renewable contract replicates an indefinite employment contract by setting the profit threshold level in such a way that even a manager that competes in the product market will achieve that profit level, i.e., $\underline{\pi} = 0$. A short-term contract then effectively mimics the collusive stability associated with a long-term contract. In that case, a *strict* short-term, renewable contract with $\underline{\pi} > 0$ would actually destabilize collusion.

3.4 Comparative Statics

The only assumptions on profit distributions are full support over the domain $\pi_t \geq 0$ and first order stochastic dominance of defection over collusion over Nash. Therefore, any change in profit distributions has an ambiguous impact on the *direction* as well as the *size* of the change of both $E_a(\pi_t)$ and $G_a(\underline{\pi})$. The ultimate impact on Proposition 1's condition (5) then depends on the relative changes in $E_a(\pi_t)$ and $G_a(\underline{\pi})$.

As a result, even very stylized comparative statics with respect to profit distributions leads to a large number of cases entailing tedious analyses without generating substantive insights in the model. The key point remains that changes in the parameters facilitate collusive stability if (i) the increase in reappointment probability during defection becomes smaller; (ii) the reduction in reappointment probability during punishment becomes larger; or (iii) the monetary loss from punishment becomes smaller—see the first paragraph after Proposition 1.

However, since the distribution over profit when defecting only appears once in condition (5) in the form of $G_D(\underline{\pi})$, some meaningful comparative statics are possible.

Proposition 3 Cond. (5) is easier (more difficult) to satisfy if distribution $F_D(x)$ is

- 1. shifted to the left (right);
- 2. more (less) dispersed, provided that $\underline{\pi} < E_D(\pi_t)$ and keeping $E_D(\pi_t)$ fixed; or
- 3. less (more) dispersed, provided that $\underline{\pi} \geq E_D(\pi_t)$ and keeping $E_D(\pi_t)$ fixed.

The intuition is that changes entailing a reduction (increase) in $G_D(\underline{\pi})$, while not affecting the other parameters, make condition (5) easier (more difficult) to satisfy, because such changes make defection more attractive to the manager in the sense that the probability of re-employment increases after a defection.

4 Fixed Salary Components and Serial Colluders

This section considers the impact on cartel stability of fixed salary components (4.1), as well as the possibility to revert from punishment to collusion after replacement of a manager, which I refer to as "serial collusion" (4.2).

4.1 Fixed Salary Components Can Stabilize Cartels

This subsection relaxes the assumption that contracts contain no fixed salary component, that is, the shareholders are now free to set any $\alpha \ge 0$ and $\beta \ge 0$. This shows that a fixed salary component affects cartel stability when short-term contracts are in place, but not when long-term contracts are in place.

Benchmark stability. First, consider long-term contracts. With long-term contracts, there exists a rather trivial optimal solution in which $\beta = 0$, because then the manager is indifferent between all possible actions as he is hired forever and earns α in every period. He has no incentives to defect, which makes collusion stable for all discount factors. This solution is not so interesting; therefore, I assume $\beta > 0$ from now on.

When $\beta > 0$ the stability with long-term contracts is unchanged when allowing for a fixed salary component, because collusion is stable if and only if

$$\frac{\delta}{1-\delta} \left[\left(\alpha + \beta E_C \left(\pi_t \right) \right) - \left(\alpha + \beta E_N \left(\pi_t \right) \right) \right] \ge \left(\alpha + \beta E_D \left(\pi_t \right) \right) - \left(\alpha + \beta E_N \left(\pi_t \right) \right),$$

iff.
$$\frac{\delta}{1-\delta} \left(E_C \left(\pi_t \right) - E_N \left(\pi_t \right) \right) \ge E_D \left(\pi_t \right) - E_C \left(\pi_t \right),$$
 (7)

which is equivalent to benchmark stability condition (2). The reason is that the fixed salary component is paid out in every period and, thus, cancels out.

Stability with short-term, renewable contracts. The following lemma states the stability condition with short-term contracts and a fixed salary component.

Lemma 2 Allowing for a fixed salary component α , collusion is stable if and only if

$$\delta \left(\frac{G_C(\underline{\pi})}{1 - \delta G_C(\underline{\pi})} E_C(\pi_t) - \frac{G_D(\underline{\pi})}{1 - \delta G_N(\underline{\pi})} E_N(\pi_t) + K \frac{\alpha}{\beta} \right) \ge$$

$$E_D(\pi_t) - E_C(\pi_t), \qquad (8)$$

where $K = \frac{G_C(\underline{\pi})}{1 - \delta G_C(\underline{\pi})} - \frac{G_D(\underline{\pi})}{1 - \delta G_N(\underline{\pi})}$.

Proof. Collusion is stable if and only if the discounted expected payoff from defection is not larger than the discounted expected payoff from collusion, i.e., if and only if

$$\delta\left(\frac{G_{C}\left(\underline{\pi}\right)}{1-\delta G_{C}\left(\underline{\pi}\right)}\left(\alpha+\beta E_{C}\left(\pi_{t}\right)\right)-\frac{G_{D}\left(\underline{\pi}\right)}{1-\delta G_{N}\left(\underline{\pi}\right)}\left(\alpha+\beta E_{N}\left(\pi_{t}\right)\right)\right)\geq \alpha+\beta E_{D}\left(\pi_{t}\right)-\left(\alpha+\beta E_{C}\left(\pi_{t}\right)\right),$$

which gives condition (8) when rearranging terms.

For now, note that this stability condition differs from the stability condition without the fixed salary component in Lemma 1 by the term $\delta K_{\beta}^{\alpha}$ on the LHS. I provide intuition for this difference below. Defining $\tilde{\delta} = \frac{G_D(\pi) - G_C(\pi)}{G_C(\pi)(G_D(\pi) - G_N(\pi))}$, the following lemma states the optimal salary scheme from the shareholder's perspective and the impact of the fixed salary component α on the stability of collusion.

Lemma 3 If $\delta > \tilde{\delta}$, collusion is stable and the optimal wage schedule entails an arbitrarily small fraction of firm profit $\beta > 0$ and a positive fixed salary component $\alpha > 0$. If $\delta \leq \tilde{\delta}$, collusion is stable if and only if condition (3) holds and the optimal wage schedule entails an arbitrarily small fraction of firm profit $\beta > 0$ and a zero fixed salary component $\alpha = 0$.

Proof. We have that $K > 0 \Leftrightarrow \delta > \tilde{\delta}$. Therefore, if $\delta > \tilde{\delta}$, the LHS of condition (8) can be made arbitrarily large by setting α to any positive value and β arbitrarily small. However, if $\delta \leq \tilde{\delta}$, it is best to set $\alpha = 0$ (because K < 0), and collusion is stable if and only if condition (3) holds.

The intuition for this result runs as follows. Dividing all managerial payoffs by β , we have that with action $a \in \{N, C, D\}$, the manager's expected payoff is $\frac{\alpha}{\beta} + E_a(\pi_t)$. The normalized fixed component $\frac{\alpha}{\beta}$ cancels out in "the gain from defection"—i.e., the RHS of condition (8) because the fixed component is being paid in the current period independent of managerial behavior, that is, independent of whether the manager colludes or defects. However, the fixed component does not cancel out in the "future losses from punishment"—i.e., the LHS of condition (8)—because the manager is being paid the fixed component in a future period if and only if the manager is actually hired in that future period, the probability of which depends on whether the manager colludes or defects in the current period.

If the "future losses from punishment with regard to the fixed component" is positive, i.e., if K > 0, then it is optimal for the shareholder to set $\alpha > 0$ so as to use the fixed salary component to increase stability. The shareholder can amplify this effect by making the fixed component a relatively important part of the wage schedule compared to the variable component, that is, by setting β sufficiently small relative to α . Note that if $\delta > \tilde{\delta}$, stability does not depend on the size of $\underline{\pi}$, because the shareholder can play around with α and β to make collusion stable for every $\underline{\pi}$. The next Proposition summarizes these results.

Proposition 4 Allowing for a salary scheme with a fixed component weakly stabilizes collusion with short-term contracts, while not affecting collusive stability with long-term contracts.

Proof. If $\tilde{\delta}$ satisfies condition (3), then we know from Lemma 3 that it is optimal to set $\alpha = 0$, resulting in condition (3) and (8) to be equivalent. If $\tilde{\delta}$ violates condition (3), then we know from Lemma 3 that it is optimal to set $\alpha > 0$, resulting in condition (8) to hold for a larger set of discount factor than condition (3). From (7) we know that a fixed salary component does not affect collusive stability.

4.2 Serial Colluders Destabilize Cartels

This subsection assumes that managers revert from punishment to collusion after one of the managers has been replaced during the punishment phase. The motivation is that mistrust may be eliminated when a new manager is appointed; when a new manager enters the firm, he starts with a "clean sheet" and will restore collusion with his rival.

Proposition 5 If replacement of a manager during the punishment regime results in the newly appointed manager to switch to collusion with his rival, then collusion is stable for less discount factors than if collusion is not restored after a managerial replacement.

Proof. Let $V_C = \frac{E_C(\pi_t)}{1-\delta G_C(\underline{\pi})}$ be the manager's continuation value of the collusive state. When collusion cannot (can) be restored, let V_D (V_D^r) and V_N (V_N^r) be the continuation value of the punishment state and defection state state, respectively. From (1), $G_T(\underline{\pi}) \leq G_C(\underline{\pi})$ is the probability of hitting at least profit level $\underline{\pi}$ when the rival manager defected from the collusive agreement.

Without restoration of collusion, collusion is stable if and only if $V_C \ge V_D$, where

$$V_D = E_D(\pi_t) + \delta G_D(\underline{\pi}) V_N.$$

With restoration of collusion, collusion is stable if and only if $V_C \ge V_D^r$, where

$$V_D^r = E_D(\pi_t) + \delta G_D(\underline{\pi}) \left[G_T(\underline{\pi}) V_N^r + (1 - G_T(\underline{\pi})) V_C \right]$$

and V_N^r is determined by solving

$$V_N^r = E_N(\pi_t) + \delta G_N(\underline{\pi}) [G_N(\underline{\pi}) V_N^r + (1 - G_N(\underline{\pi})) V_C],$$

$$\Leftrightarrow V_N^r = \frac{E_N(\pi_t) + \delta G_N(\underline{\pi}) (1 - G_N(\underline{\pi})) V_C}{1 - \delta (G_N(\underline{\pi}))^2}.$$

Noting that $V_N = \frac{E_N(\pi_t)}{1 - \delta G_N(\pi)}$, straightforward algebra yields $V_N^r > V_N$, and, therefore, we have $V_D^r > V_D$.

If a manager can switch from punishment to collusion with a newly appointed rival manager, this effectively results in the continuation value of defection to increase for two reasons. First, defection results in an immediate reduction in the probability that the rival manager is reappointed, that is, an increase in the probability that a *new* rival manager is appointed and collusion is restored instantaneously.¹⁷ Second, the continuation value of defection is higher as in each punishment period collusion will be restored with probability $G_N(\underline{\pi})$. In essence, Proposition 5 echoes McCutcheon (1997) who argues that meetings facilitate renegotiation, which effectively reduces the scope for punishment and thereby destabilizes collusion.

5 Short-Term Contracts Entail Equilibrium Price Wars

Typically, CEO contracts span several years.¹⁸ Therefore, this section studies the model when managers interact *multiple* times on the product market *within* the course of their contractual employment period. This allows for dynamic pricing in the sense that a manager may find it optimal to make his behavior during his employment period contingent on profit realizations occurring earlier within the same contractual period.

5.1 Set-up of the Dynamic Model

To keep the analysis clean and to focus on the impact of multiple interactions within the same employment period on cartel behavior, managers are assumed to interact on the product market *twice* during their employment period, and that they do *not* discount their payoffs

¹⁷In the extreme case in which $G_T = 0$, collusion is unstable for every discount factor $\delta \in [0, 1]$ as defection automatically results in a new rival manager being appointed immediately: collusion is restored instantaneously and no punishment occurs.

¹⁸See, for example, Schwab and Thomas (2006) and Gillian, Hartzell and Parrino (2009).

within the employment period. More than two interactions within the same contractual period and within-contract discounting would not bring essential insights, while mathematically complicating the model substantially.

Consider the model as outlined in Subsection 3.1, but now suppose that each period $t \in \{1, 2, ..., \infty\}$ consists of two sequential stages $k \in \{1, 2\}$. In both stages, the managers take action pair $a_k \in \{N, C, D\}$, resulting in profit $\pi_t^k \ge 0$, and the manager is reappointed for the next period t + 1 if and only if $\pi_t^1 + \pi_t^2 \ge \underline{\pi}$. The conditional probability distributions over profit are the same as outlined in Subsection 3.1 and independent between periods and stages. Also, as before, managerial behavior is observed by the rival manager immediately after taking the action, thus allowing for punishment during the next interaction. The new timing of the game is depicted in Figure 3.

In terms of notation, define $G_{a_1a_2}(\underline{\pi}) = \Pr[\pi_t^1 + \pi_t^2 \ge \underline{\pi} | a_1, a_2]$ as the *ex ante* probability of realizing at least aggregate profit $\underline{\pi}$ in period t, provided that the managers take action pair a_1 in stage 1 and a_2 in stage 2. Noting that $E_a(\pi_t^1) = E_a(\pi_t^2)$, I leave out the argument π_t^k in the expected value $E_a(\pi_t^k)$ for ease of notation, which then simply writes as E_a .

To derive more precise results, assume the following regularity condition,

$$\frac{\partial \left(G_D\left(x\right) - G_C\left(x\right)\right)}{\partial x} > 0,\tag{9}$$

that is, the difference in re-employment probability between defection and collusion increases in the profit threshold $\underline{\pi}$. This means that increasing the profit threshold $\underline{\pi}$ reduces the probability of re-employment more when colluding than when defecting.



Figure 3. Timing of the dynamic game.

5.2 Collusion in Both Stages

The most straightforward grim-trigger collusive strategy is to collude in both stages as long as the rival manager colluded in all previous stages in all previous periods, while punishing forever otherwise. The stability of this strategy is pinned down by the following lemma. Lemma 4 Collusion in both stages is a stable strategy if and only if

$$2\delta\left(\frac{G_C(\underline{\pi})}{1-\delta G_{CC}(\underline{\pi})}E_C - \frac{G_D(\underline{\pi})}{1-\delta G_{NN}(\underline{\pi})}E_N\right) \ge E_D - E_C.$$
(10)

Proof. See Appendix B.

For a collusive strategy to be stable in the dynamic model, the manager must (i) not defect in stage 1, and (ii) not defect in stage 2 after any possible profit realization in stage 1, $\pi_t^1 \geq 0$. Consider first the managerial incentive to defect in stage 2. The monetary tradeoff is independent of the profit realization in stage 1: defection immediately increases the manager's expected salary from E_C to E_D , while decreasing it from E_C to E_N in each future punishment stage. However, the re-employment tradeoff does depend on the profit realization in stage 1: defection in stage 2 immediately increases the probability of re-employment from G_C ($\pi - \pi_t^1$) to G_D ($\pi - \pi_t^1$), while decreasing it from G_{NN} (π) to G_{CC} (π) in each future punishment period. Since defection increases the immediate re-employment probability more the higher is the profit level that needs to be attained (by the regularity condition), the manager finds it most attractive to defect when the profit realization in the first stage was the lowest realization possible—i.e., $\pi_t^1 = 0$ —because then he must hit a high profit of at least $\underline{\pi} - \pi_t^1 = \underline{\pi}$ in the second stage, the probability of which is substantially increased when defecting.

Consider now the managerial incentive to defect in stage 1. This is lower than the managerial incentive to defect in stage 2 after the realization of profit $\pi_t^1 = 0$. The reason is two-fold. First, in stage 1, the manager has two interactions to attain aggregate profit $\pi_t^1 + \pi_t^2 \ge \underline{\pi}$, while after the realization of a zero profit in stage 1 the manager has only one interaction left to attain $\pi_t^2 \ge \underline{\pi}$. Second, defection in stage 1 is immediately followed by punishment in stage 2 without intra-period discounting, while defection in stage 2 is followed by punishment in the next period, which entails discounting.

Combining the arguments, collusion in both stages is a stable strategy if and only if the manager is patient enough not to defect in the second stage after the realization of a zero profit in the first stage, which is represented by condition (10).

5.3 Collusion Conditional on Profit Realization in Stage 1

The dynamic nature of the game allows the manager to condition his action in the second stage on the profit realization in the first stage. By the argument based on the regularity condition in the previous subsection, if the manager is patient enough to collude in stage 2 after profit realization $\pi_t^1 = \tau$, then he is also patient enough to collude in stage 2 after profit realization $\pi_t^1 > \tau$, but may not be patient enough to collude in stage 2 after any profit realization $\pi_t^1 < \tau$. The intuition is that the closer is stage 1's profit realization to the profit threshold $\underline{\pi}$, the less attractive is defection in terms of increasing the immediate re-employment probability.

Compared to collusion in both stages, the manager potentially increases cartel stability by adopting a strategy that entails collusion in stage 2 only after profit realization $\pi_t^1 \ge \tau$, which I define as τ -conditional collusion. The reason is that the stability of collusion in both periods is determined by the manager's incentive to defect in stage 2 after a zero profit realization in stage 1, while τ -conditional collusion removes this concern as the manager competes as part of the collusive strategy for all $\pi_t^1 < \tau$. The next two subsections investigate the stability of τ -conditional collusion in stage 2 when the manager either colludes or competes in stage 1.

Collusion in stage 1 and conditional collusion in stage 2. Consider the collusive strategy in which the manager colludes in stage 1, while colluding in stage 2 if and only if $\pi_t^1 \geq \tau$. Define the *ex ante* re-employment probability of this strategy as $P_C(\underline{\pi}, \tau) = \int_{\tau}^{\infty} \Pr(\pi_t^1 = x | a = C) G_C(\underline{\pi} - x) dx + \int_0^{\tau} \Pr(\pi_t^1 = x | a = C) G_N(\underline{\pi} - x) dx$, & denote the *ex ante* expected per-period payoff as $K_C(\tau) = (1 + G_C(\tau)) E_C + (1 - G_C(\tau)) E_N$. The following lemma pins down the stability of this strategy.

Lemma 5 Collusion in stage 1 and τ -cond. collusion in stage 2 is a stable strategy iff.

$$G_{C}(\tau) \left(E_{C} - E_{N}\right) + \delta\left(\frac{P_{C}(\underline{\pi}, \tau)}{1 - \delta P_{C}(\underline{\pi}, \tau)} K_{C}(\tau) - \frac{G_{DN}(\underline{\pi})}{1 - \delta G_{NN}(\underline{\pi})} 2E_{N}\right) \geq E_{D} - E_{C}, \text{ and}$$
(11)

$$\delta\left(\frac{G_C\left(\underline{\pi}-\tau\right)}{1-\delta P_C\left(\underline{\pi},\tau\right)}K_C\left(\tau\right) - \frac{G_D\left(\underline{\pi}-\tau\right)}{1-\delta G_{NN}\left(\underline{\pi}\right)}2E_N\right) \ge E_D - E_C,\tag{12}$$

and is potentially more stable than collusion in both stages.

Proof. See Appendix C.

By the regularity condition, the lower is stage 1's profit realization, the more attractive is defection in stage 2. Thus, the critical constraint for the manager's equilibrium behavior in stage 2 to be stable is determined by the lowest possible profit realization in stage 1 that still entails collusion in stage 2, that is, profit realization $\pi_t^1 = \tau$. This constraint is represented by condition (12): τ -conditional collusion in stage 2 (i) potentially makes collusion more stable than sure collusion in stage 2 as defection entails a lower increase in re-employment probability, that is $G_D(\underline{\pi} - \tau) - G_C(\underline{\pi} - \tau) < G_D(\underline{\pi}) - G_C(\underline{\pi})$, but (ii) may also decrease collusive stability as the equilibrium payoff is lower than collusion in both stages, that is $K_C(\tau) < 2E_C$, and its equilibrium continuation probability is lower, that is $P_C(\underline{\pi}, \tau) < G_{CC}(\underline{\pi})$. The net effect depends on the precise specification of the density functions over profit and the level of τ .

If the manager is patient enough to collude in stage 2 after $\pi_t^1 \ge \tau$, he is not necessarily also patient enough to collude in stage 1. The intuition is that in stage 1, the manager has two interactions to achieve aggregate profit $\pi_t^1 + \pi_t^2 \ge \underline{\pi}$, while in stage 2 the binding constraint prescribes that the manager has one interaction to achieve profit $\pi_t^2 \ge \underline{\pi} - \tau$. Depending on the precise specification of the density functions over profit and the level of τ , the constraint in stage 1, i.e., (11), may or may not be the binding constraint.

Competition in stage 1 and conditional collusion in stage 2. If constraint (11) is the binding constraint for collusion in stage 1 and τ -conditional collusion in stage 2 to be stable, the manager may circumvent this and increase stability by constructing a collusive agreement that entails competition in stage 1 and τ -conditional collusion in stage 2. Define the *ex ante* re-employment probability associated with this strategy as $P_N(\underline{\pi}, \tau) = \int_{\tau}^{\infty} \Pr(\pi_t^1 = x | a = N) G_C(\underline{\pi} - x) dx + \int_0^{\tau} \Pr(\pi_t^1 = x | a = N) G_N(\underline{\pi} - x) dx$, and denote the *ex ante* expected per-period payoff as $K_N(\tau) = G_N(\tau) E_C + (2 - G_N(\tau) E_N)$. The following lemma pins down the stability of this strategy.

Lemma 6 Competition in stage 1 and τ -cond. collusion in stage 2 is a stable strategy iff.

$$\delta\left(\frac{G_C\left(\underline{\pi}-\tau\right)}{1-\delta P_N\left(\underline{\pi},\tau\right)}K_N\left(\tau\right) - \frac{G_D\left(\underline{\pi}-\tau\right)}{1-\delta G_{NN}\left(\underline{\pi}\right)}2E_N\right) \ge E_D - E_C,\tag{13}$$

and is potentially more stable than (i) collusion in both stages, and (ii) collusion in stage 1 and τ -conditional collusion in stage 2.

Proof. See Appendix D.

Stability is not a concern in the first stage, simply because the collusive strategy prescribes competition in stage 1. Similarly to the argument presented in the previous subsection, the binding constraint for the collusive strategy to be stable in the second stage is determined by profit realization $\pi_t^1 = \tau$. If constraint (11) is the binding constraint in Lemma 5, then competition in stage 1 instead of collusion in stage 1 increases cartel stability provided that constraint (13) is satisfied for a larger set of discount factors than constraint (11), which depends on the precise specification of the density functions over profit and the level of τ .

5.4 Equilibrium Price Wars, Stability, and Profitability

This subsection collects the results from the previous three lemmas and summarizes the behavioral implications when the managers adopt a strategy with τ -conditional collusion in stage 2.

Proposition 6 Short-term contracts spanning multiple managerial interactions on the product market can entail price wars in equilibrium, thereby increasing cartel stability, while reducing cartel profitability.

Proof. See Appendix E.

Consider a manager colluding in both periods. Then, the critical constraint for collusion to be stable is whether the manager is patient enough to collude in stage 2 after a low profit realization in stage 1, because the manager has only one opportunity left to attain its contractual profit threshold level. However, if this jeopardizes cartel stability, then the managers may fix stability by adopting a collusive strategy in which they always collude in the first stage, while colluding in the second stage only if profit realization in the first stage was relatively high. In that way, they eliminate the concern of defection after a low profit realization, because the collusive equilibrium itself already specifies competition after a low profit realization.

It may be the case that such a strategy is still not stable, because the managers have incentives to defect in the first stage now. Then, the managers can change their strategy and compete in stage 1 as part of the collusive equilibrium, while colluding in stage 2 only if the profit realization in stage 1 is high enough.

Both strategies entail competition (price wars) as part of the collusive equilibrium. Although such price wars increase cartel stability, they decrease cartel profitability, because by stochastic dominance (i) expected profit of competition is lower than expected profit of collusion, and (ii) the continuation probability of collusion in both stages is higher than the continuation probability when the manager competes in the second (and possibly: the first) stage with some probability.

6 How Debt-Financed Firms Can Form Stable Cartels

The model can also be interpreted in light of firm financing, thereby showing how firms that are financed by debt, i.e., highly leveraged firms, can form more-stable cartels than firms financed by equity. This finding is consistent with Schleifer and Vishny (1992) and Khanna and Tice (2000), although through a different mechanism. In the spirit of the model presented in this paper, consider the stylized situation of a firm fully financed by short-term debt. Assume that (i) the bank liquidates this firm immediately if it misses out on a periodical repayment, and (ii) all the profits left after the repayment are reinvested in the firm. Thus, to meet its repayment obligations, this debt-financed firm must each year make a profit of at least the amount that needs to be repaid. If such a firm colludes with its rivals, it faces the following tradeoff: defection makes it more likely to meet its current repayment obligation (due to a higher expected profit in the current period), but punishment makes it less likely to meet its future payment obligations (due to lower expected profits in future punishment periods). In other words, defection makes liquidation less likely in the current period, but more likely in future periods. This parallels the managerial tradeoff that defection makes being fired less likely in the current period, but more likely in future periods. Therefore, debt contracts can be designed in such a way to stabilize collusion and may be a source of cyclical pricing.¹⁹

7 Concluding Remarks

This paper has presented a mechanism through which commonly observed short-term, renewable employment contracts can improve cartel stability, while dynamically affecting firms' pricing behavior. Motivated by empirical observations, the model takes such short-term, renewable contracts as exogenously given, thereby abstracting away from the question which type of employment contracts would endogenously arise when the shareholders' main goal is to induce the CEO to either form or stay away from a cartel. This issue is at the center of the analysis in Aubert (2009) and Angelucci and Han (2011).

The model assumes that managers can observe each other's actions. However, if managers would not be able to observe the rival's action, Green and Porter (1984) suggest that the collusive strategy entails reversion to Nash for some periods after a low profit realization. How such unobservability of actions interacts with the impact of short-term, renewable contracts on cartel stability is left for future research.

In the current model, the shareholder's strategy to hire a new manager when profit turns out to be low is rational, because hiring costs are assumed to be zero and all managers are of the same type. It is then costless to hire a new manager. In contrast, if hiring costs are positive,²⁰ one may expect the current manager to haggle with the shareholder to renew his contract after a low profit realization; this is work for future research.

 $^{^{19}}$ Apart from the described forces, another important force is that liquidation of the firm will result in a change of market structure, which would *ex ante* affect incentives. The impact of this force on overall cartel stability is an interesting avenue to explore in future research.

²⁰This may be due to, for example, search costs or costs associated with drafting a new contract.

The model treats a long-term contract as an infinite contract such that the manager is never dismissed, thereby implicitly assuming that the termination cost of a long-term contract is infinitely high. However, sometimes a long-term contract can be terminated by paying some fixed firing cost, such as a "golden parachute". In such cases, a long-term contract may display the same collusion-stabilizing effect as a short-term contract as the contract is then effectively not indefinite.

Finally, the mechanism outlined in this paper implicitly assumes that employment contracts are observable to rival managers. Although recently, more and more firms publish the remuneration package and employment conditions of their executives, this may not always be the case. When rivals' employment contracts are not observable, then correct beliefs about these contracts may still induce the cartel-stabilizing mechanism.

Appendix

A Proof of Proposition 3

(1) Shifting $F_D(x)$ to the left (right) increases (decreases) $G_D(\underline{\pi})$, which in turn increases (decreases) the LHS of condition (5). (2) Suppose $\underline{\pi} < E_D(\pi_t)$. Making $F_D(x)$ more (less) dispersed increases (decreases) $G_D(\underline{\pi})$, which in turn increases (decreases) the LHS of condition (5). (3) Suppose $\underline{\pi} \ge E_D(\pi_t)$. Making $F_D(x)$ more (less) dispersed decreases (increases) $G_D(\underline{\pi})$, which in turn decreases (increases) the LHS of condition (5).

B Proof of Lemma 4

The manager has no incentive to defect in stage 1 if and only if

$$\frac{2E_C}{1 - \delta G_{CC}\left(\underline{\pi}\right)} \ge E_D + E_N + \delta G_{DN}\left(\underline{\pi}\right) \frac{2E_N}{1 - \delta G_{NN}\left(\underline{\pi}\right)},$$

while, given the realization of π_t^1 , he has no incentive to defect in stage 2 if and only if

$$E_C + \delta G_C \left(\underline{\pi} - \pi_t^1\right) \frac{2E_C}{1 - \delta G_{CC} \left(\underline{\pi}\right)} \ge E_D + \delta G_D \left(\underline{\pi} - \pi_t^1\right) \frac{2E_N}{1 - \delta G_{NN} \left(\underline{\pi}\right)}$$

These two conditions rewrite as

$$\min\left\{E_C - E_N + 2\delta\left(\frac{G_{CC}\left(\underline{\pi}\right)}{1 - \delta G_{CC}\left(\underline{\pi}\right)}E_C - \frac{G_{DN}\left(\underline{\pi}\right)}{1 - \delta G_{NN}\left(\underline{\pi}\right)}E_N\right);\tag{14}$$

$$2\delta\left(\frac{G_C\left(\underline{\pi}-\pi_t^1\right)}{1-\delta G_{CC}\left(\underline{\pi}\right)}E_C - \frac{G_D\left(\underline{\pi}-\pi_t^1\right)}{1-\delta G_{NN}\left(\underline{\pi}\right)}E_N\right)\right\} \ge E_D - E_C \tag{15}$$

for every profit realization $\pi_t^1 \ge 0$. Subcondition (15) can be rewritten as

$$2\delta \left[\frac{G_C \left(\underline{\pi} - \pi_t^1\right) - G_D \left(\underline{\pi} - \pi_t^1\right)}{1 - \delta G_{NN} \left(\underline{\pi}\right)} E_N + G_C \left(\underline{\pi} - \pi_t^1\right) \left(\frac{E_C}{1 - \delta G_{CC} \left(\underline{\pi}\right)} - \frac{E_N}{1 - \delta G_{NN} \left(\underline{\pi}\right)} \right) \right] \ge E_D - E_C,$$

where we note that (i) $G_C(\underline{\pi} - \pi_t^1)$ is increasing in π_t^1 , and (ii) $G_C(\underline{\pi} - \pi_t^1) - G_D \times (\underline{\pi} - \pi_t^1)$ is also increasing in π_t^1 by regularity condition (9). Thus, Subcondition (15) is most difficult to satisfy if $\pi_t^1 = 0$, that is, if

$$2\delta \left(\frac{G_C(\underline{\pi})}{1 - \delta G_{CC}(\underline{\pi})} E_C - \frac{G_D(\underline{\pi})}{1 - \delta G_{NN}(\underline{\pi})} E_N \right) \ge E_D - E_C.$$
(16)

Is (16) more difficult to satisfy than Subcondition (14)? Subtracting the LHS of (14) from the LHS of (16) yields

$$A := E_C - E_N + 2\delta \left(\frac{G_{CC}\left(\underline{\pi}\right) - G_C\left(\underline{\pi}\right)}{1 - \delta G_{CC}\left(\underline{\pi}\right)} E_C - \frac{G_{DN}\left(\underline{\pi}\right) - G_D\left(\underline{\pi}\right)}{1 - \delta G_{NN}\left(\underline{\pi}\right)} E_N \right),$$

which can be rewritten as

$$G_{CC}(\underline{\pi}) - G_C(\underline{\pi}) = \int_0^\infty \Pr\left(\pi_t^1 = x \,\middle|\, a = C\right) G_C(\underline{\pi} - x) \, dx - G_C(\underline{\pi}) \tag{17}$$

$$= \int_{0}^{\underline{\pi}} \Pr\left(\pi_{t}^{1} = x \middle| a = C\right) G_{C}\left(\underline{\pi} - x\right) dx \tag{18}$$

$$+\int_{\underline{\pi}} \Pr\left(\pi_{t}^{1} = x \mid a = C\right) G_{C}\left(\underline{\pi} - x\right) dx - G_{C}\left(\underline{\pi}\right)$$

$$= \int_{0}^{\underline{\pi}} \Pr\left(\pi_{t}^{1} = x \mid a = C\right) G_{C}\left(\underline{\pi} - x\right) dx \qquad (19)$$

$$+ \int_{\underline{\pi}}^{\infty} \Pr\left(\pi_{t}^{1} = x \mid a = C\right) dx - G_{C}\left(\underline{\pi}\right)$$

$$= \int_{0}^{\underline{\pi}} \Pr\left(\pi_{t}^{1} = x \mid a = C\right) G_{C}\left(\underline{\pi} - x\right) dx + G_{C}\left(\underline{\pi}\right)$$

$$- G_{C}\left(\underline{\pi}\right) \qquad (20)$$

$$= \int_0^{\underline{\pi}} \Pr\left(\pi_t^1 = x \,\middle|\, a = C\right) G_C\left(\underline{\pi} - x\right) dx,\tag{21}$$

where (17) follows by definition, (18) follows by splitting up the integral, (19) follows by noting that $G_C(\underline{\pi} - x) = 1$ for every $x \in [\underline{\pi}, \infty)$, (20) follows by definition, yielding (21). Similarly,

$$G_{DN}(\underline{\pi}) - G_D(\underline{\pi}) = \int_0^\infty \Pr\left(\pi_t^1 = x \middle| a = D\right) G_N(\underline{\pi} - x) \, dx - G_D(\underline{\pi})$$
$$= \int_0^{\underline{\pi}} \Pr\left(\pi_t^1 = x \middle| a = D\right) G_N(\underline{\pi} - x) \, dx + G_D(\underline{\pi})$$
$$- G_D(\underline{\pi}).$$

Therefore, $G_{CC}(\underline{\pi}) - G_C(\underline{\pi}) \ge G_{DN}(\underline{\pi}) - G_D(\underline{\pi})$, because by stochastic dominance we have (i) $\int_0^{\underline{\pi}} \Pr(\pi_t^1 = x | a = C) dx \ge \int_0^{\underline{\pi}} \Pr(\pi_t^1 = x | a = D) dx$, and (ii) $G_C(\underline{\pi} - x) \ge G_N(\underline{\pi} - x)$ for every $x \in [0, \underline{\pi}]$. Thus, noting that $E_C > E_N$ and $\frac{E_C}{1 - \delta G_{CC}(\underline{\pi})} > \frac{E_N}{1 - \delta G_{NN}(\underline{\pi})}$, we have A > 0and, thus, (16) is the binding constraint.

C Proof of Lemma 5

The manager has no incentive to defect in stage 1 if and only if

$$\frac{E_C + G_C(\tau) E_C + (1 - G_C(\tau)) E_N}{1 - \delta P_C(\underline{\pi}, \tau)} \ge E_D + E_N + \delta G_{DN}(\underline{\pi}) \frac{2E_N}{1 - \delta G_{NN}(\underline{\pi})}$$

where $P_C(\underline{\pi}, \tau) = \int_{\tau}^{\infty} \Pr(\pi_t^1 = x | a = C) G_C(\underline{\pi} - x) dx + \int_0^{\tau} \Pr(\pi_t^1 = x | a = C) \times G_N(\underline{\pi} - x) dx$. Given the realization of π_t^1 , he has no incentive to defect in stage 2 if and only if

$$E_{C} + \delta G_{C} \left(\underline{\pi} - \pi_{t}^{1}\right) \frac{E_{C} + G_{C} \left(\tau\right) E_{C} + \left(1 - G_{C} \left(\tau\right)\right) E_{N}}{1 - \delta P_{C} \left(\underline{\pi}, \tau\right)} \ge E_{D} + \delta G_{D} \left(\underline{\pi} - \pi_{t}^{1}\right) \times \frac{2E_{N}}{1 - \delta G_{NN} \left(\underline{\pi}\right)},$$

which rewrites as

$$\delta\left(\frac{G_C\left(\underline{\pi}-\pi_t^1\right)}{1-\delta P_C\left(\underline{\pi},\tau\right)}K_C\left(\tau\right)-\frac{G_D\left(\underline{\pi}-\pi_t^1\right)}{1-\delta G_{NN}\left(\underline{\pi}\right)}2E_N\right)\right\}\geq E_D-E_C$$

for every profit realization $\pi_t^1 \ge \tau$. By the regularity condition, this boils down to combined conditions (11) and (12) in Lemma 5.

Noting that $P(\underline{\pi}, \tau) < G_{CC}(\underline{\pi})$ and $E_C < K_C(\tau)$, we have that (11) is more difficult to satisfy than (14). However, constraint (14) is not the binding constraint for the "always collusion strategy" to be stable; constraint (15) with $\pi_t^1 = 0$ is the binding constraint. Thus, constraint (11) does not make the " τ -conditional collusion strategy" less stable than strategy "always collusion strategy" as long as τ is such that (11) is easier to satisfy than (15) with $\pi_t^1 = 0$, which depends on the precise specification of the density functions.

Fixing $\pi_t^1 \geq 0$, constraint (12) is more difficult to satisfy than constraint (15), because $P(\underline{\pi}, \tau) < G_{CC}(\underline{\pi})$ and $E_C < K_C(\tau)$. However, depending on the precise specification of the density functions, constraint (12) may be easier to satisfy than constraint (15), because (i) both constraints are more difficult to satisfy the lower is π_t^1 ; and (ii) constraint (15) needs to be satisfied for every $\pi_t^1 \geq 0$; while (iii) constraint (12) needs only to be satisfied for every $\pi_t^1 \geq \tau$.

Thus, depending on the precise specification of the density functions, choosing an appropriate τ potentially results in both constraints (11) and (12) to be satisfied for a larger set of discount factors than constraint (15).

D Proof of Lemma 6

Given the realization of π_t^1 , the manager has no incentive to defect in stage 2 if and only if

$$E_{C} + \delta G_{C} \left(\underline{\pi} - \pi_{t}^{1}\right) \frac{E_{N} + G_{N} \left(\tau\right) E_{C} + \left(1 - G_{N} \left(\tau\right)\right) E_{N}}{1 - \delta P_{N} \left(\underline{\pi}, \tau\right)} \ge E_{D} + \delta G_{D} \left(\underline{\pi} - \pi_{t}^{1}\right) \times \frac{2E_{N}}{1 - \delta G_{NN} \left(\underline{\pi}\right)},$$

which rewrites as

$$\delta\left(\frac{G_C\left(\underline{\pi}-\pi_t^1\right)}{1-\delta P_N\left(\underline{\pi},\tau\right)}K_N\left(\tau\right)-\frac{G_D\left(\underline{\pi}-\pi_t^1\right)}{1-\delta G_{NN}\left(\underline{\pi}\right)}2E_N\right)\geq E_D-E_C,$$

for every profit realization $\pi_t^1 \ge \tau$. By the regularity condition, we have that the above constraint is most difficult to satisfy if $\pi_t^1 = \tau$; thus, stability is determined by condition (13) in Lemma 6.

E Proof of Proposition 6

First claim: price wars in equilibrium. The collusive strategies described in Lemmas 5 and 6 entail competition in stage 2 after the realization of profit $\pi_t^1 < \tau$ in stage 1; also, the collusive strategy described in Lemma 6 always entails competition in stage 1. Moreover, the collusive strategies described in Lemmas 5 and 6 are potentially more stable than collusion in both stages. Thus, contracts that span multiple managerial interactions potentially entail price wars in equilibrium.

Second claim: if managers adopt a strategy entailing price wars in equilibrium, then cartel stability is increased. The collusive strategies described in Lemmas 5 and 6 are potentially more stable, but entail a lower profitability—see the proof below. Thus, if managers adopt such a strategy, it means that they are not patient enough to collude in both stages, and thus adopt a strategy entailing equilibrium price wars, while compromising on cartel profitability.

Third claim: price wars reduce cartel profitability. The profitability of (i) collusion in both stages; (ii) collusion in stage 1 and τ -conditional collusion in stage 2; and (iii) competition in stage 1 and τ -conditional collusion in stage 2 is, respectively,

$$\frac{2E_{C}}{1-\delta G_{CC}\left(\underline{\pi}\right)} > \frac{K_{C}\left(\tau\right)}{1-\delta P_{C}\left(\underline{\pi},\tau\right)} > \frac{K_{N}\left(\tau\right)}{1-\delta P_{N}\left(\underline{\pi},\tau\right)}$$

because $2E_C > K_C(\tau) > K_N(\tau)$ and $G_{CC}(\underline{\pi}) > P_C(\underline{\pi},\tau) > P_N(\underline{\pi},\tau)$.

References

- Abreu, D., Pearce, D. and E. Stacchetti. (1986). Optimal Cartel Equilibria With Imperfect Monitoring. Journal of Economic Theory, 39, 251–269.
- Angelucci, C. and M.A. Han. (2011). Monitoring Managers Through Corporate Compliance Programs. Amsterdam Center for Law Economics (ACLE) Working Paper No. 2010-15.
- Aubert, C. (2009). Managerial Effort Incentives and Market Collusion. Toulouse School of Economics Working Paper No. 09-127.
- Aubert, C., Rey, P. and W.E. Kovacic. (2006). The Impact of Leniency and Whistle-Blowing Programs on Cartels. *International Journal of Industrial Organization*, 24, 1241–1266.
- Baker, G., Gibbons, R. and K.J. Murphy. (200). Relational Contracts and the Theory of the Firm. *The Quarterly Journal of Economics*, 117, 39–84.
- Bolton, P., Scheinkman, J. and W. Xiong. (2006). Executive Compensation and Short-Termist Behavior in Speculative Markets. *Review of Economic Studies*, 73, 577–610.
- Buccirossi, P. and G. Spagnolo. (2007). Corporate Governance and Collusive Behavior. In D.W. Collins (Ed.), *Issues in Competition Law and Policy*. Chicago: American Bar Association.
- Bull, C. (1987). The Existence of Self-Enforcing Implicit Contracts. The Quarterly Journal of Economics, 102, 147–159
- Chen, Z. (2008). Cartel Organization and Antitrust Enforcement. University of East Anglia CCP Working Paper No. 08-21.
- Farrell, K.A. and D.A. Whidbee. (2003). Impact of Firm Performance Expectations on CEO Turnover and Replacement Decisions. *Journal of Accounting and Economics*, 36, 165–196.
- Fershtman, C. (1985). Internal Organizations and Managerial Incentives as Strategic Variables in Competitive Environment. International Journal of Industrial Organization, 3, 245–253.
- Fershtman, C. and K.L. Judd. (1987). Equilibrium Incentives in Oligopoly. The American Economic Review, 77, 927–940.
- Fershtman, C. and A. Pakes. (2000). A Dynamic Oligopoly With Collusion and Price Wars. RAND Journal of Economics, 31, 207–236.
- Gillian, S.L., Hartzell, J.C. and R. Parrino. (2009). Explicit vs. Implicit Contracts: Evidence from CEO Employment Agreements. *Journal of Finance*, 64, 1629–1655.

- Green, E.J. and R.H. Porter. (1984). Noncooperative Collusion Under Imperfect Price Information. *Econometrica*, 52, 87–100.
- Haltiwanger, J. and J.E. Harrington Jr. (1991). The Impact of Cyclical Demand Movements on Collusive Behavior. RAND Journal of Economics, 22, 89–106.
- Han, M.A. (2011). Strategic Delegation Improves Cartel Stability. Amsterdam Center for Law Economics (ACLE) Working Paper No. 2010-14.
- Harrington Jr., J.E. (2004a). Cartel Pricing Dynamics in the Presence of an Antitrust Autority. *RAND Journal of Economics*, 35, 651–673.
- Harrington Jr., J.E. (2004b). Post-Cartel Pricing During Litigation. Journal of Industrial Economics, 52, 517–533.
- Harrington Jr., J.E. (2006). How do Cartels Operate? Foundations and Trends in Microeconomics, 2, 1–105.
- Harrington Jr., J.E. and J. Chen. (2006). Cartel Pricing Dynamics With Cost Variability and Endogenous Buyer Detection. *International Journal of Industrial Organization*, 24, 1185–1212.
- Jenter, D. and F. Kanaan. (2010). CEO Turnover and Relative Performance Evaluation. Journal of Finance (Forthcoming).
- Jenter, D. and K. Lewellen. (2010). Performance-Induced CEO Turnover. Working Paper Stanford.
- Kaplan, S.N. and B.A. Minton. (2010). How Has CEO Turnover Changed? NBER Working Paper No. W12465.
- Khanna, N. and S. Tice. (2000). Strategic Responses of Incumbents to New Entry: The Effect of Ownership Structure, Capital Structure, and Focus. *Review of Financial Studies*, 13, 749–779.
- Lambertini, L. and M. Trombetta. (2002). Delegation and Firms' Ability to Collude. Journal of Economic Behavior & Organization, 47, 359–373.
- Lausten, M. (2002). CEO Turnover, Firm Performance and Corporate Governance: Empirical Evidence on Danish Firms. International Journal of Industrial Organization, 20, 391–414.
- Levenstein, M.C. (1997). Price Wars and the Stability of Collusion: A Study of the Pre-World War I Bromine Industry. Journal of Industrial Economics, 45, 117–137.

- Levenstein, M.C., and V.Y. Suslow. (2006). What Determines Cartel Success? Journal of Economic Literature, 44, 43–95.
- McCutcheon, B. (1997). Do Meetings in Smoke-Filled Rooms Facilitate Collusion? The Journal of Political Economy, 105, 330–350.
- Neubecker, L. (2005). Collusion and Cyclic Pricing by Managers in Markets with Fluctuating Demand. *Economics Letters*, 88, 164–169.
- Rotemberg, J.J. and G. Saloner. (1986). A Supergame-Theoretic Model of Price Wars during Booms. The American Economic Review, 76, 390–407.
- Schleifer, A. and R.W. Vishny. (1992). Liquidation Values and Debt Capacity: A Market Equilibrium Approach. *The Journal of Finance*, 47, 1343–1366.
- Schwab, S.J. and R.S. Thomas (2006). An Empirical Analysis of CEOF Employment Contracts: What Do Top Executives Bargain For? Washington and Lee Law Review, 63, 231–270.
- Sklivas, S.D. (1987). The Strategic Choice of Managerial Incentives. RAND Journal of Economics, 18, 452–458.
- Slade, M.E. (1989). Price Wars in Price-Setting Supergames. *Economica*, 56, 295–310.
- Slade, M.E. (1990). Strategic Pricing Models and Interpretation of Price-War Data. European Economic Review, 34, 524–537.
- Spagnolo, G. (2000). Stock-Related Compensation and Product-Market Competition. RAND Journal of Economics, 31, 22–42.
- Spagnolo, G. (2005). Managerial Incentives and Collusive Behavior. European Economic Review, 49, 1501–1523.
- Stigler, G.J. (1964). A Theory of Oligopoly. The Journal of Political Economy, 72, 44-61.
- Vickers, J. (1985). Delegation and the Theory of the Firm. *Economic Journal*, 95, 138–147.

SFB 649 Discussion Paper Series 2012

For a complete list of Discussion Papers published by the SFB 649, please visit http://sfb649.wiwi.hu-berlin.de.

- 001 "HMM in dynamic HAC models" by Wolfgang Karl Härdle, Ostap Okhrin and Weining Wang, January 2012.
- 002 "Dynamic Activity Analysis Model Based Win-Win Development Forecasting Under the Environmental Regulation in China" by Shiyi Chen and Wolfgang Karl Härdle, January 2012.
- 003 "A Donsker Theorem for Lévy Measures" by Richard Nickl and Markus Reiß, January 2012.
- 004 "Computational Statistics (Journal)" by Wolfgang Karl Härdle, Yuichi Mori and Jürgen Symanzik, January 2012.
- 005 "Implementing quotas in university admissions: An experimental analysis" by Sebastian Braun, Nadja Dwenger, Dorothea Kübler and Alexander Westkamp, January 2012.
- 006 "Quantile Regression in Risk Calibration" by Shih-Kang Chao, Wolfgang Karl Härdle and Weining Wang, January 2012.
- 007 "Total Work and Gender: Facts and Possible Explanations" by Michael Burda, Daniel S. Hamermesh and Philippe Weil, February 2012.
- 008 "Does Basel II Pillar 3 Risk Exposure Data help to Identify Risky Banks?" by Ralf Sabiwalsky, February 2012.
- 009 "Comparability Effects of Mandatory IFRS Adoption" by Stefano Cascino and Joachim Gassen, February 2012.
- 010 "Fair Value Reclassifications of Financial Assets during the Financial Crisis" by Jannis Bischof, Ulf Brüggemann and Holger Daske, February 2012.
- 011 "Intended and unintended consequences of mandatory IFRS adoption: A review of extant evidence and suggestions for future research" by Ulf Brüggemann, Jörg-Markus Hitz and Thorsten Sellhorn, February 2012.
- 012 "Confidence sets in nonparametric calibration of exponential Lévy models" by Jakob Söhl, February 2012.
- 013 "The Polarization of Employment in German Local Labor Markets" by Charlotte Senftleben and Hanna Wielandt, February 2012.
- 014 "On the Dark Side of the Market: Identifying and Analyzing Hidden Order Placements" by Nikolaus Hautsch and Ruihong Huang, February 2012.
- 015 "Existence and Uniqueness of Perturbation Solutions to DSGE Models" by Hong Lan and Alexander Meyer-Gohde, February 2012.
- 016 "Nonparametric adaptive estimation of linear functionals for low frequency observed Lévy processes" by Johanna Kappus, February 2012.
- 017 "Option calibration of exponential Lévy models: Implementation and empirical results" by Jakob Söhl und Mathias Trabs, February 2012.
- 018 "Managerial Overconfidence and Corporate Risk Management" by Tim R. Adam, Chitru S. Fernando and Evgenia Golubeva, February 2012.
- 019 "Why Do Firms Engage in Selective Hedging?" by Tim R. Adam, Chitru S. Fernando and Jesus M. Salas, February 2012.
- 020 "A Slab in the Face: Building Quality and Neighborhood Effects" by Rainer Schulz and Martin Wersing, February 2012.
- 021 "A Strategy Perspective on the Performance Relevance of the CFO" by Andreas Venus and Andreas Engelen, February 2012.
- 022 "Assessing the Anchoring of Inflation Expectations" by Till Strohsal and Lars Winkelmann, February 2012.

SFB 649, Spandauer Straße 1, D-10178 Berlin http://sfb649.wiwi.hu-berlin.de



This research was supported by the Deutsche Forschungsgemeinschaft through the SFB 649 "Economic Risk".

SFB 649 Discussion Paper Series 2012

For a complete list of Discussion Papers published by the SFB 649, please visit http://sfb649.wiwi.hu-berlin.de.

- 023 "Hidden Liquidity: Determinants and Impact" by Gökhan Cebiroglu and Ulrich Horst, March 2012.
- 024 "Bye Bye, G.I. The Impact of the U.S. Military Drawdown on Local German Labor Markets" by Jan Peter aus dem Moore and Alexandra Spitz-Oener, March 2012.
- 025 "Is socially responsible investing just screening? Evidence from mutual funds" by Markus Hirschberger, Ralph E. Steuer, Sebastian Utz and Maximilian Wimmer, March 2012.
- 026 "Explaining regional unemployment differences in Germany: a spatial panel data analysis" by Franziska Lottmann, March 2012.
- 027 "Forecast based Pricing of Weather Derivatives" by Wolfgang Karl Härdle, Brenda López-Cabrera and Matthias Ritter, March 2012.
- ⁰²⁸ "Does umbrella branding really work? Investigating cross-category brand loyalty" by Nadja Silberhorn and Lutz Hildebrandt, April 2012.
- 029 "Statistical Modelling of Temperature Risk" by Zografia Anastasiadou, and Brenda López-Cabrera, April 2012.
- 030 "Support Vector Machines with Evolutionary Feature Selection for Default Prediction" by Wolfgang Karl Härdle, Dedy Dwi Prastyo and Christian Hafner, April 2012.
- 031 "Local Adaptive Multiplicative Error Models for High-Frequency Forecasts" by Wolfgang Karl Härdle, Nikolaus Hautsch and Andrija Mihoci, April 2012.
- 032 "Copula Dynamics in CDOs." by Barbara Choroś-Tomczyk, Wolfgang Karl Härdle and Ludger Overbeck, May 2012.
- 033 "Simultaneous Statistical Inference in Dynamic Factor Models" by Thorsten Dickhaus, May 2012.
- "Realized Copula" by Matthias R. Fengler and Ostap Okhrin, Mai 2012.
- 035 "Correlated Trades and Herd Behavior in the Stock Market" by Simon Jurkatis, Stephanie Kremer and Dieter Nautz, May 2012
- ⁰³⁶ "Hierarchical Archimedean Copulae: The HAC Package" by Ostap Okhrin and Alexander Ristig, May 2012.
- ⁰³⁷ "Do Japanese Stock Prices Reflect Macro Fundamentals?" by Wenjuan Chen and Anton Velinov, May 2012.
- 038 "The Aging Investor: Insights from Neuroeconomics" by Peter N. C. Mohr and Hauke R. Heekeren, May 2012.
- 039 "Volatility of price indices for heterogeneous goods" by Fabian Y.R.P. Bocart and Christian M. Hafner, May 2012.
- 040 "Location, location: Extracting location value from house prices" by Jens Kolbe, Rainer Schulz, Martin Wersing and Axel Werwatz, May 2012.
- 041 "Multiple point hypothesis test problems and effective numbers of tests" by Thorsten Dickhaus and Jens Stange, June 2012
- 042 "Generated Covariates in Nonparametric Estimation: A Short Review." by Enno Mammen, Christoph Rothe, and Melanie Schienle, June 2012.
- 043 "The Signal of Volatility" by Till Strohsal and Enzo Weber, June 2012.
- 044 "Copula-Based Dynamic Conditional Correlation Multiplicative Error Processes" by Taras Bodnar and Nikolaus Hautsch, July 2012

MD H. SITA, Y

SFB 649, Spandauer Straße 1, D-10178 Berlin http://sfb649.wiwi.hu-berlin.de

This research was supported by the Deutsche Forschungsgemeinschaft through the SFB 649 "Economic Risk".

SFB 649 Discussion Paper Series 2012

For a complete list of Discussion Papers published by the SFB 649, please visit http://sfb649.wiwi.hu-berlin.de.

- 045 "Additive Models: Extensions and Related Models." by Enno Mammen, Byeong U. Park and Melanie Schienle, July 2012.
- 046 "A uniform central limit theorem and efficiency for deconvolution estimators" by Jakob Söhl and Mathias Trabs, July 2012
- 047 "Nonparametric Kernel Density Estimation Near the Boundary" by Peter Malec and Melanie Schienle, August 2012
- 048 "Yield Curve Modeling and Forecasting using Semiparametric Factor Dynamics" by Wolfgang Karl Härdle and Piotr Majer, August 2012
- 049 "Simultaneous test procedures in terms of p-value copulae" by Thorsten Dickhaus and Jakob Gierl, August 2012
- 050 "Do Natural Resource Sectors Rely Less on External Finance than Manufacturing Sectors?" by Christian Hattendorff, August 2012
- 051 "Using transfer entropy to measure information flows between financial markets" by Thomas Dimpfl and Franziska J. Peter, August 2012
- 052 "Rethinking stock market integration: Globalization, valuation and convergence" by Pui Sun Tam and Pui I Tam, August 2012
- 053 "Financial Network Systemic Risk Contributions" by Nikolaus Hautsch, Julia Schaumburg and Melanie Schienle, August 2012
- 054 "Modeling Time-Varying Dependencies between Positive-Valued High-Frequency Time Series" by Nikolaus Hautsch, Ostap Okhrin and Alexander Ristig, September 2012
- 055 "Consumer Standards as a Strategic Device to Mitigate Ratchet Effects in Dynamic Regulation" by Raffaele Fiocco and Roland Strausz, September 2012
- 056 "Strategic Delegation Improves Cartel Stability" by Martijn A. Han, October 2012
- 057 "Short-Term Managerial Contracts and Cartels" by Martijn A. Han, October 2012



SFB 649, Spandauer Straße 1, D-10178 Berlin http://sfb649.wiwi.hu-berlin.de